Internal control and financial reporting quality of small firms

A comparative analysis of regulatory regimes

Jagan Krishnan and Jayanthi Krishnan Department of Accounting, Temple University, Philadelphia, Pennsylvania, USA, and

Sophie Liang School of Management, Binghamton University, Binghamton, New York, USA

Abstract

Purpose – The Dodd–Frank Act of 2010 exempts small, non-accelerated filers from compliance with Sarbanes–Oxley Act (SOX) Section 404b internal control audits. However, these firms are required to comply with other internal control regulations, namely, SOX Sections 302 and 404a, starting in 2002 and 2007, respectively. A small number of these firms also voluntarily adopted (and sometimes dropped) Section 404b during 2004-2010. The purpose of this study is to investigate the impact of a series of internal control regulations introduced by SOX on the financial reporting quality of small firms.

Design/methodology/approach – The research design for this study is empirical. Using unsigned and signed discretionary accruals as measures of financial reporting quality, the authors compare the financial reporting quality for adopters and non-adopters across four regulation regimes over the period 2000-2010: PRESOX, SOX 302, SOX 404a and SOX 404b.

Findings – The results indicate that most of the adopters and non-adopters benefited from SOX 302 and 404a compared with the PRESOX period. However, only the non-adopters gained incrementally when moving from SOX 302 to SOX 404a. Also, Section 404b benefited firms with material weaknesses, as well as firms without material weaknesses that had the lowest reporting quality, in the PRESOX period.

Research limitations/implications – This study helps inform the important policy debate on whether to increase the threshold that is used for the SOX 404b exemption. It shows incremental benefits for firms that adopted Section 404b audits, even when they were complying with Section 302 and Section 404a. Consequently, extending the exemption to more companies would result in a loss of the reporting quality benefit of 404b.

Originality/value – This study contributes to the literature by focusing exclusively on non-accelerated filers and by examining differences across four regulation regimes over a long window compared to prior studies. It provides evidence that the financial reporting benefit of SOX 404b is not transitional, but rather extends for a few years even after some firms discontinued the 404b audits.

Keywords Dodd-Frank Act, SOX Section 302, SOX Section 404, Material weaknesses

Paper type Research paper



The authors appreciate the comments of two anonymous reviewers, Jennifer Ho (editor), Matt Ege, Hyun Jong Park, Padmakumar Sivadasan, and workshop participants at AAA Auditing Section Midyear Meeting, AAA Annual Meeting, and Temple University. Professor Jagan Krishnan acknowledges research support from Temple University's Sabbatical, and Professor Sophie L. Liang acknowledges summer research funding from the School of Management at Binghamton University.

Review of Accounting and Finance Vol. 19 No. 2, 2020 pp. 221-246 © Emerald Publishing Limited 1475-7702 DOI 10.1108/RAF-05-2018-0107



 $\mathbf{221}$

Received 23 May 2018 Revised 7 September 2019 13 February 2020 Accepted 17 February 2020

RAF 1. Introduction

19.2

222

The Sarbanes–Oxley Act (SOX) introduced multiple overlapping regulations relating to companies' internal controls. Section 302 of the Act requires management assertions about the effectiveness of disclosure controls. Section 404a requires management reports, based on appropriate documentation and testing, on the effectiveness of internal control over financial reporting (ICFR), and Section 404b requires auditor reports on their clients' internal controls. The regulations were designed to enhance the reliability of financial reporting and to improve investor confidence. In this study, we investigate the impact of a series of internal control regulations introduced by SOX on the financial reporting quality of small firms.

Our investigation is particularly salient because, after several postponements of the effective date for non-accelerated filers to comply with Section 404b, the Dodd–Frank Wall Street Reform and Consumer Protection Act of 2010 (henceforth Dodd–Frank Act) permanently exempted non-accelerated filers (i.e. companies with public float less \$75m) from 404b. There has been controversy over Section 404b ICFR audits, revolving around the assertion that its costs outweigh its benefits, particularly for smaller firms. However, the exemption has also been controversial, with critics arguing that it will result in different levels of protections for investors, "confuse investors and may undermine their confidence in financial reporting" (SEC, 2011)[1].

In this paper, we focus specifically on the exempted firms – non-accelerated filers – and examine trends in their financial reporting quality for the period 2000-2010. Of particular interest to us, a small number of these firms voluntarily adopted Section 404b during this period, while a majority mandatorily adopted Section 302 in 2002 and Section 404a in 2007. This provides us with the opportunity to examine the effects of the different policies, which are associated with increasing costs, as companies move from the PRESOX period to Section 302, Section 404a and Section 404b (Kinney and Shepardson, 2011).

We address the following two research questions. First, did non-accelerated filers benefit differentially from Sections 302, 404a and 404b, and more specifically, were there *incremental* benefits resulting from Section 404b for those that voluntarily adopted it? We compare the financial reporting quality of the latter firms (the "adopters") before, during and, for some who discontinued these audits, after the period in which the audits were conducted, with that of firms who did not adopt Section 404b (the "non-adopters").

The second research question is, do the benefits from Section 404b, if any, accrue across the board to all firms? In particular, we examine whether the results differ across "clean" firms that had no material weaknesses throughout our sample period, and "internal control weakness (ICW)" firms, which had some internal control weaknesses during our sample period. This question is important because of the perception that any benefits from Section 404b accrue only to the relatively small proportion of firms that have internal control problems.

Previous research has examined the potential costs of the Dodd–Frank Act exemption by comparing the financial reporting quality of non-accelerated filers that were not subject to SOX 404b with that of small accelerated filers that were subject to it (Nagy, 2010; Krishnan and Yu, 2012; Holder *et al.*, 2013). Similarly Kinney and Shepardson (2011) examine the "relative performance" of Sections 404a and 404b regulations using the disclosure rates of material weaknesses for non-accelerated filers and small accelerated filers as the measure of benefits. Unlike these studies, our study contributes to the literature by focusing exclusively on non-accelerated filers, and by examining differences across four regulation regimes. Unlike Kinney and Shepardson (2011), we use discretionary accruals as a more general measure to capture the financial reporting quality benefits of the regulations. Although the internal control regulations are intended to identify material weaknesses, the broader goal



(stemming from the goal of SOX), stated repeatedly by the Securities and Exchange Commission (SEC) is "improving investor confidence in the reliability of a company's financial disclosure and system of internal control over financial reporting." The process of evaluation can help early identification of potential weaknesses, facilitating the "continuous, orderly and timely flow of information within the company and, ultimately, to investors and the marketplace" and early detection of fraudulent financial reporting (SEC, 2003). If the processes involved in the execution of the regulations engenders improvements in overall reporting – even for firms with no reported weaknesses – we would expect to capture it with our measure[2]. Our study also contributes to the literature by examining the impact of SOX 404b audits over a longer window than prior studies (Kinney and Shepardson, 2011; Holder *et al.*, 2013). SEC former Chairman William Donaldson states that despite its large short-term costs, SOX 404 may have "the greatest long-term potential to improve financial reporting by public companies" (Donaldson, 2005). Our evidence suggests that the financial reporting benefit of SOX 404b is not transitional, but rather, it extends for a few years even after some firms discontinued the 404b audits.

Our sample consists of firm-year observations for non-accelerated filers for the period 2000-2010. We compare the financial reporting quality for adopters and non-adopters across four regulation regimes: PRESOX, SOX 302, SOX 404a and SOX 404b. It is important to note that these regimes have multiple regulations. The SOX 404a regime includes Sections 302 and 404a, both of which require management reports that have significant overlaps. The SOX 404b regime includes Sections 302, 404a and 404b.

We use both absolute and signed discretionary accruals as our measure of financial reporting quality. While prior studies (Jones, 1991; Teoh *et al.*, 1998) use signed discretionary accruals to test for earnings management in a particular direction, many recent studies use absolute discretionary accruals to examine the magnitude of earnings management because it captures both upward and downward management. Our full sample results indicate, using both accruals measures, that both adopters and non-adopters benefited from SOX 302 and 404a compared with the PRESOX period. Further, the adopters gained incrementally over SOX 404a when they moved to SOX 404b audits.

To address our second research question, we divide our sample into clean and ICW firms. We find that the benefits of SOX 302 – the least costly regulation alternative – accrue to both clean and ICW firms. However, we see no benefits to either group when SOX 404a was superimposed on SOX 302. Finally, SOX 404b brought benefits only to adopters with material weaknesses. We further examined whether the results for the clean firms differed across levels of PRESOX quality, since firms with ex ante high quality of reporting may not benefit from internal control regulation. We find that a majority of the firms had lower accruals in the SOX 302 period compared with the PRESOX period. However, benefits from SOX 404b were confined to clean firms in the highest quintile of discretionary accruals in the PRESOX period. Overall, we conclude that, in addition to firms with internal control weaknesses, clean firms with the lowest starting reporting quality benefited from SOX 404b audits.

By focusing our analyses of the incremental benefit of SOX 404b audits completely on non-accelerated filers, we contribute to the continuing debate on the threshold that should be used for the SOX 404b exemption. After the passage of the Dodd–Frank Act, legislators continue to consider expanding the exemption to more firms[3]. In May 2019, the SEC proposed amendments to filer definitions that will allow more companies to claim exemption from Section 404b audits (SEC, 2019). The proposed changes have been controversial, even within the SEC. Proponents of expanding exemption, including SEC Chairman Jay Clayton, noted that Section 404b adds significant costs but is unlikely to enhance reporting quality



Financial reporting quality of small firms

(Clayton, 2019). Commissioner Elad L. Roisman questioned whether benefits of Section 404b outweigh the costs for the firms in question, especially when they must still comply with Section 302 and Section 404a (Roisman, 2019). Commissioner Robert Jackson Jr. voted against the amendments, arguing that they would exempt high-growth firms "where the risk, and consequences, of fraud are greatest – where the benefits of the auditor's presence are highest" (Jackson, 2019). Our findings help inform this important policy debate as we show the incremental benefits for firms that adopted Section 404b audits, even when they were complying with Section 302 and Section 404a. Consequently, extending the exemption to more companies would reduce compliance costs but also result in the loss of the reporting quality benefits of 404b.

The rest of the paper is organized as follows. Section 2 describes the regulatory background and research relating to internal controls. Section 3 describes our research questions and research design. Section 4 describes our sample and results. Section 5 provides our conclusions.

2. Background and related research

2.1 Internal control regulation

RAF

19.2

224

The landscape pertaining to internal control disclosure and audits expanded significantly with the passage of the SOX. SOX introduced two major sections relating to internal control. Section 302, which became effective on August 29, 2002, requires a company's CEO and CFO to provide quarterly certifications about their evaluation of disclosure controls (Hermanson and Ye, 2009). Section 404 comprises two parts, Sections 404a and 404b. Section 404a requires management of companies to provide an assessment of internal control and Section 404b requires their auditors to provide an independent assessment of the effectiveness of their clients' ICFR.

Effective dates for Section 404 were set differently for accelerated filers (companies with public float greater than or equal to \$75m) and non-accelerated files (companies with public float less \$75m)[4]. Accelerated filers were required to comply with Sections 404a and 404b starting for fiscal years ending on or after November 15, 2004 (SEC, 2005). For non-accelerated filers, Section 404a became effective for fiscal years ending on or after December 15, 2007. However, the effective date for non-accelerated filers to comply with the requirements of Section 404b (internal control audits) was set initially for fiscal years ending on or after April 15, 2005, postponed several times, and set finally for fiscal years ending June 15, 2010. However, the Dodd–Frank Wall Street Reform and Consumer Protection Act of 2010 (henceforth Dodd–Frank Act) permanently exempted non-accelerated filers from compliance with Section 404b.

The exemption decision has been controversial. Supporters of the Dodd–Frank Act argue that it protects non-accelerated filers from the potentially high compliance costs of Section 404b[5]. Others argue that there has been a decreasing trend in the cost of compliance with Section 404b, partly because of firms and auditors gaining compliance experience, and partly to the changes to the ICFR process introduced by Auditing Standard 5, which replaced the previous standard (Auditing Standard 2) in 2007 (SEC, 2011). In addition, many believe that compliance with Section 404b has strengthened investor protection and increased investor confidence, and reduced the incidence of financial restatements (SEC, 2011).

Even though non-accelerated filers were never required to comply with 404b, some companies adopted ICFR audits *voluntarily* during the years 2004-2010 in anticipation of mandatory compliance with Section 404b. Following the multiple extensions of the effective dates for complying with Section 404b, and the passage of the Dodd–Frank Act, some of



these non-accelerated filers that had voluntarily adopted ICFR audits decided to discontinue them, possibly due to the high costs of compliance. The availability of voluntary adopters provides us with a unique natural experiment to explore the link between ICFR audits and financial reporting quality.

2.2 Related literature

The large literature on internal control that has emerged since the passage of SOX has some notable features. First, a vast majority of these studies focus on the determinants and consequences of material weaknesses. Because of this focus, several studies combine disclosures from Sections 302 and 404 (and samples using both non-accelerated and accelerated filers), where the latter are Section 404b auditor-discovered weaknesses. Second, again because of the focus on material weaknesses, very few studies (discussed below) have examined the general effects of the regulations on the financial reporting quality of "clean" companies that do not have weaknesses. Third, very few studies present comparisons of different regulation regimes, and to our knowledge, no study has compared the financial reporting quality effects of the Sections 302 and 404a disclosures.

2.2.1 Studies focused on ICW disclosure and difference across regimes[6]. Studies on the determinants of ICWs show that they tend to arise in firms that are small, financially distressed, with complex and fast-changing operations, without sufficient commitment of resources for accounting controls or undergoing restructuring (Doyle *et al.*, 2007a). Several studies document negative associations between material weaknesses and financial reporting quality proxies, such as discretionary accruals, accruals quality, restatements, accounting conservatism, and earnings persistence (Doyle *et al.*, 2007b; Ashbaugh-Skaife *et al.*, 2008; Goh and Li, 2011). Finally, several studies examine the market response to the disclosures of ICWs (Hammersley *et al.*, 2008; Beneish *et al.*, 2008) and the association of ICWs with debt contracting features (Costello and Wittenberg-Moerman, 2011; Kim *et al.*, 2011), cost of equity and cost of debt (Ogneva *et al.*, 2007; Beneish *et al.*, 2008; Ashbaugh-Skaife *et al.*, 2009; Dhaliwal *et al.*, 2011; Kim *et al.*, 2011; Gordon and Wilford, 2012).

While many studies combine data for both Sections 302 and 404, a few studies provide comparisons of the two sections. Doyle *et al.* (2007b) document that Section 302 ICW disclosures are more strongly associated with lower accruals quality than 404 disclosures. Ashbaugh-Skaife *et al.* (2008) find that firms that remediate internal control deficiencies show improvement in accruals quality compared to those that do not report remediation. Corporate governance characteristics such as the quality of audit committee and the board are associated with the existence of 404 disclosures of ICWs, but not 302 disclosures (Hoitash *et al.*, 2009). Beneish *et al.* (2008) document negative market reactions to Section 302 disclosures but not to Section 404 disclosures. Hermanson and Ye (2009) focus on the accelerated filers with adverse initial 404b reports and find that only 27 per cent of the companies provided early warning of material weaknesses through 302 disclosures.

Using the disclosure rates for ICWs as the measure of the benefits from the regulations, Kinney and Shepardson (2011) find similar increases in the disclosure rates for nonaccelerated filers providing first-time Section 404a reports and for small accelerated filers providing first-time Section 404b reports. Kinney and Shepardson conclude that the two regimes yield similar benefits, whereas they impose very dissimilar costs. Our study differs from Kinney and Shepardson (2011) in that we examine how a series of internal control regulations affect firms over the years, while they focus on the change in disclosure rates during the first year after an internal control regulation is implemented. In addition, Kinney and Shepardson (2011) measure quality with material weakness disclosure rates,



Financial reporting quality of small firms

RAF 19.2

226

which implicitly assumes that the underlying quality does not change over time. We employ a more general measure of financial reporting quality that does not require such assumption. This allows us to examine potential benefits of the regulation for a more general population of firms, not just those with material weaknesses.

2.2.2 Other studies on benefits of IC regulation. While all of the above studies have focused on firms with ICWs, two studies examine a general population of firms, those with and without ICWs. Chen *et al.* (2013) examine the informativeness of clean SOX 404 internal control opinions. They find that the earnings informativeness for accelerated filers was higher for clean internal control reports in the SOX 404 adoption year compared to the previous years. In contrast, there was no change in the earnings informativeness for non-accelerated filers. Using a sample of non-accelerated filers that voluntarily underwent SOX 404 internal control audits, Cassell *et al.* (2013) document a significantly lower cost of debt and equity for voluntary adopters, especially in the year of adoption compared to those that did not have internal controls audits.

Three other studies (Nagy, 2010; Krishnan and Yu, 2012; Holder *et al.*, 2013) compare small accelerated filers and non-accelerated filers to examine whether non-accelerated filers will benefit from the requirement of internal control audits. Nagy (2010) provides evidence of a negative association between companies that underwent SOX 404 audits and restatements. Holder *et al.* (2013) find that the reporting quality of non-accelerated filers compared to that of accelerated filers declined in the post-SOX period. Krishnan and Yu (2012) find that investors perceive small public firms that underwent SOX 404 audits as having higher revenue and earnings quality, relative to firms that are not subject to 404 audits. All of these studies use two groups of companies, small accelerated filers and non-accelerated filers, to conduct their analyses. We focus specifically on the exempted firms – non-accelerated filers – and examine trends in their financial reporting quality over the years.

3. Hypotheses development and research design

We divide our sample period, 2000-2010, into five sub-periods for non-accelerated filers (Figure 1), four indicating "regulation regimes" and one period for filers that dropped 404b compliance:

- PRESOX: 2000-2001;
- SOX 302: period during which a firm was only responsible for Section 302 reports;
- SOX 404a: period during which a firm was only responsible for Sections 302 and 404a reports;
- SOX 404b: period during which a firm was responsible for Sections 302 and 404a, but also adopted Section 404b *voluntarily*; and
- AFTERDROP: period after which a firm discontinued Section 404b audits.

Figure 1 depicts these regimes. We distinguish between firms that voluntarily adopted Section404b ("adopters") and those that did not ("non-adopters") during the period 2004-2010. Note that, while the first three regimes were required for all non-accelerated filers, the fourth regime is applicable only to the adopters. The actual years for each regime above (following PRESOX) differ for non-adopters and adopters. For non-adopters, the SOX 302 and SOX 404a regimes cover 2002-2006, and 2007-2010 respectively. For adopters however, two possibilities exist. Some adopted Section 404b before 2007 and some after 2007. Thus for those that adopted Section 404b before 2007, there are only three regimes (PRESOX, SOX 302 and SOX 404b). Firms that adopted Section 404b in or after 2007 have all four regimes listed above. In addition to periods corresponding to the regulation regimes, some adopters





Notes: $_{a,b}$ There are two possibilities for adopters. Some firms adopted Section 404b before 2007 and some after 2007. Thus, for those that adopted Section 404b before 2007, there are only three regulation regimes (PRESOX, SOX 302 and SOX 404b). Firms that adopted Section 404b in or after 2007 have all four regulation regimes (PRESOX, SOX 302, SOX 404a and SOX 404b). $_{c,d,e}$ In addition to periods corresponding to the regulation regimes, some adopters have a period (AFTERDROP) because they dropped their ICFR audits (Section 404b) after a few years

have an additional period (AFTERDROP) because they dropped their ICFR audits (Section 404b) after a few years. As Figure 1 shows, not all adopters went through all five periods. Some adopters, for example, adopted Section 404b before 2007, and do not have a separate 404a period. Others did not discontinue Section 404b until 2010 and therefore do not have a post404b period.

3.1 Research questions and hypotheses

The distinction between the regulation regimes is important because they impose different costs on the complying firms. Specifically, the effort required of management and auditors, and therefore the total compliance costs, increase from PRESOX to SOX 302, SOX 302 to SOX 404a, and from SOX 404a to SOX 404b (Kinney and Shepardson, 2011). Do the benefits to financial reporting quality also increase as we move from the least to most costly regulation regime? We pose our first research question as follows:

RQ1. Did non-accelerated filers benefit differentially from Sections 302, 404a and 404b?

As we discuss below, while we expect that SOX 302, SOX 404a and SOX 404b would yield incremental financial reporting quality benefits over the PRESOX regime, we do not predict an increasing trend of benefits similar to those discussed for compliance costs above.

Section 302 requires reports on companies' certifying officers' evaluation of the effectiveness of "disclosure controls and procedures." The SEC expects such evaluations to result in "[...] significant benefits by ensuring that information about an issuer's business and financial condition is adequately reviewed by the issuer's principal



www

Figure 1.

Regulation regimes

RAF
19,2
executive and financial officers [...]" Although this suggests that the SOX 302 regime would yield financial reporting quality benefits over the previous PRESOX regime, such an outcome is not certain because the rule did not provide specific guideline or procedures for management[7]. Lack of clear guidance could also mean that not all material weaknesses were discovered (Hermanson and Ye, 2009; Chen *et al.*, 2013), and therefore any benefits to overall financial reporting quality could vary across firms. Nevertheless, we state the following hypothesis:

H1a. Financial reporting quality is higher in the SOX 302 regime compared with the PRESOX regime.

Unlike Section 302, Section 404a requires "high effort management ICFR reports" (Kinney and Shepardson, 2011). These are based on evaluations of the design of internal controls as well as on tests of controls. Moreover, the company must "[...] maintain evidential matter, including documentation, to provide reasonable support for management's assessment of the effectiveness of the company's internal control over financial reporting" (SEC, 2003). A priori therefore, we expect the SOX 404a regime to yield a higher degree of financial reporting quality than the SOX 302 regime. This is consistent with Kinney and Shepardson (2011)'s prediction that disclosure rates of material weaknesses by non-accelerated filers in the first year of Section 404a reports would be higher than in the previous (Section 302) year. We hypothesize the following:

H1b. Financial reporting quality is higher in the SOX 404a regime compared with the SOX 302 regime.

Section 404b requires auditor reports on the effectiveness of their clients' internal control, following procedures described in PCAOB's Auditing Standard 2 and later (after 2007) in Auditing Standard 5 (Doogar *et al.*, 2010). The procedures require the auditor to obtain "[...] sufficient competent evidence about the design and operating effectiveness of controls over all relevant financial statement assertions related to all significant accounts and disclosures in the financial statements" [Public Company Accounting Oversight Board (PCAOB), 2004]. Moreover, the internal control and financial statement audits are to be conducted in an integrated manner. A priori we would expect that the ICFR audits would improve financial reporting quality beyond the Section 302 reporting. Based on reported material weakness rates, Kinney and Shepardson (2011) argue that Section 404b is not incrementally beneficial over Section 404a. However, studies comparing non-accelerated filers with small accelerated filers conclude that the latter (with SOX 404b reporting) have higher reporting quality than the former (with only SOX 404a reporting). We hypothesize the following:

- *H1c.* Financial reporting quality is higher in the SOX 404b regime compared with the SOX 302 regime for non-accelerated filers that voluntarily adopted SOX 404b.
- *H1d.* Financial reporting quality is higher in the SOX 404b regime compared with the SOX 404a regime for non-accelerated filers that voluntarily adopted SOX 404b.

We turn next to differences between clean and ICW firms.

RQ2. Is there a difference in the effect of 404b regime on the financial reporting quality of clean and ICW firms?

As discussed, most of the focus in the literature has been on disclosure of ICWs and the impact on financial reporting quality of these disclosures. Research shows that the firms



with Section 404b material weaknesses benefit from the audits because they subsequently remedy the weaknesses and improve their financial reporting quality (Goh and Li, 2011). One frequently-voiced concern about Section 404b is that it subjects all firms to high compliance costs for these policies, even though the reports reveal that a majority of firms do not have material weaknesses[8]. The SEC and PCAOB however see general benefits for all firms, frequently stating that the internal control audits have resulted in "higher quality" financial reporting" [Public Company Accounting Oversight Board (PCAOB), 2006]. An SEC survey of financial executives shows that 49 per cent of respondents felt that Section 404 improved the quality of the company's financial reporting and 48 per cent felt that it improved the company's ability to prevent fraud. Alexander et al. (2013), based on a survey of 2,901 corporate insiders, report that "the majority of respondents recognize compliance benefits, but they do not perceive these benefits to outweigh the costs, on average" particularly among smaller companies. Recent academic evidence also provides evidence of the benefits of SOX 404 for a more general population of firms. For example, Chen et al. (2013) report that earnings informativeness was higher for a more general population of companies (i.e. those with clean reports on internal control) that adopted SOX 404b. Thus, we predict that SOX 404b would bring financial reporting quality benefits to both clean and ICW firms. Whether the benefits would be greater for ICW firms compared to clean firms is an empirical question. Therefore, we present the following hypothesis in null form.

H2. The difference in financial reporting quality between the SOX 404b and SOX 404a regimes is not different for clean firms compared to firms with internal control weaknesses.

3.2 Research design

We use the following model to test our hypotheses:

$$\begin{split} \mathrm{DA}/\mathrm{ABSDA} &= \beta_0 + \beta_1 \mathrm{NONADOPT302} + \beta_2 \mathrm{NONADOPT404a} \\ &+ \beta_3 \mathrm{ADOPTPRESOX} + \beta_4 \mathrm{ADOPT302} + \beta_5 \mathrm{ADOPT404a} \\ &+ \beta_6 \mathrm{ADOPT404b} + \beta_7 \mathrm{AFTERDROP} + \beta_8 \mathrm{BIGN} + \beta_9 \mathrm{MERGER} \\ &+ \beta_{10} \mathrm{LITIGATION} + \beta_{11} \mathrm{LOSS} + \beta_{12} \mathrm{L1ACCRUAL} \\ &+ \beta_{13} \mathrm{LEVERAGE} + \beta_{14} \mathrm{MB} + \beta_{15} \mathrm{LNMVE} + \beta_{16} \mathrm{CFO} \end{split}$$

- + β_{17} SDCFO + β_{18} SDREV + $\beta_{19}\Delta$ SALES + Time-fixed effects
- $+ \varepsilon$

(1)

Appendix I contains the variable definitions. The dependent variable, discretionary accruals (DA) or absolute discretionary accruals (ABSDA), proxies for financial reporting quality (Ashbaugh *et al.*, 2003; Kothari *et al.*, 2005). DA (ABSDA) is defined as (the absolute value of) the firm's predicted discretionary total accruals minus the median discretionary total accruals from its industry- and performance-matched portfolio.

As discussed above, our sample consists of non-accelerated filers, comprising voluntary adopters and non-adopters of Section 404b. We assign the firm-year observations for adopters to five periods, PRESOX, SOX 302, SOX 404a, SOX 404b and AFTERDROP. The firm-year observations for non-adopters are divided into three periods: PRESOX, SOX 302



Financial reporting quality of small firms

and SOX 404a. Correspondingly, we define indicator variables for adopters and nonadopters for each period. Equation (1) includes seven of the eight resulting indicator variables. The base group for the comparisons is firm-year observations for non-adopters in the PRESOX period.

NONADOPT302 and NONADOPT404a are indicator variables for non-adopters for the SOX 302 and SOX 404a periods, respectively. A negative and significant β_1 (β_2) implies a reduction in discretionary accruals and thus higher financial accounting quality for the non-adopters after SOX 302 (SOX 404a) became effective. ($\beta_2 - \beta_1$) captures the incremental improvement in financial accounting quality resulting from SOX 404a, compared to SOX 302. If SOX 404a leads to higher financial reporting quality, then we would expect ($\beta_2 - \beta_1$) to be negative.

ADOPTPRESOX, ADOPT302, ADOPT404a, ADOPT404b and AFTERDROP are indicator variables for the adopters during the PRESOX, 302, 404a, 404b and post-dropping periods, respectively. The coefficients β_3 through β_7 capture the differences in quality between adopters in the regulation regimes and non-adopters in the PRESOX period. If adopters have higher financial reporting quality in a regulation regime relative to the PRESOX period, then we expect ($\beta_k - \beta_3$, k = 4,5,6,7) to be negative. If the addition of each regime provides net incremental benefit in terms of improved financial reporting quality, relative to the previous regime, then we expect ($\beta_k - \beta_{k-1}$, k = 4,5,6,7) to be negative.

3.2.1 Discretionary accruals measure. Following Kothari *et al.* (2005), we calculate DA by first estimating the following regression separately for all firms available on the Compustat database for each year and two-digit standard industrial classification (SIC) code combination:

$$TA_{it} = \beta_0 + \beta_1 (1/ASSETS_{it-1}) + \beta_2 (\Delta Sales - \Delta AR)_{it} + \beta_3 PPE_{it} + \varepsilon_{it}$$
(2)

TA is total accruals, defined as net income before extraordinary items minus cash flow from operations, scaled by lagged total assets. The estimated residual from Equation (2) is the unadjusted DA. Following Ashbaugh *et al.* (2003), we control for the impact of performance on accruals by first partitioning observations into deciles based on their lagged return on assets (ROAs) within each two-digit SIC-year group. We then calculate DA (adjusted for firm performance) as the unadjusted DA minus the median (after excluding the corresponding firm-year observation from the portfolio) unadjusted DA of each portfolio[9].

3.2.2 Control variables. The variables corresponding to β_8 - β_{19} are firm specific controls based on previous research (Ashbaugh *et al.*, 2003; Doyle *et al.*, 2007b)[10]. We include an indicator variable for Big N auditors (BIGN) to control for audit quality. We also control for variables that are shown to be related to accruals based on previous research: market value of equity (LNMVE), cash flow from operations (CFO), loss (LOSS), change in sales (Δ SALES), market-to-book ratio (MB), leverage ratio (LEVERAGE), merger or acquisition activity (MERGER) and high litigation-risk industry (LITIGATION). We include lagged total accruals, L1ACCRUAL, to control for reversal of accruals. We also control for firms' fundamental operating volatility using the standard deviation of CFO (SDCFO) and sales volatility (SDREV) (Hribar and Nichols, 2007). Finally, we include year fixed effects and cluster standard errors at the firm level.

4. Sample and results

The sample selection process is shown in Table I, Panel A. We start with 19,515 firm-years from Audit Analytics for the period 2004-2010. We exclude 13,552 firm-years of accelerated filer firms and firms that switched filing status, leaving us with 5,963 firm-years for non-



RAF

19.2

	19,515 (13,552)	5,963			0 Total	$\begin{array}{ccc} & 14,142 \\ & 1,216 \\ & 1,216 \\ \end{array}$		AFTERDROP 189	ntrol audits in	Financi reportir quality small firn	ia ng of ns
					201	12 87			internal co	23	31
mations	4 7 4 11 0 11 5				0006	954 138		0X404b 239	SOX 404b		
ondo rear of	II-y cal UDSC	106,617 (17,056) (23,939) (16,770)	$\begin{array}{c} 48,852\\20,051\\(90)\\(4,603)\end{array}$	15,358	8006	1,064 143	c	ת	ly adopted		
# of firm					2002	1,220 120	vations	.04a .6 4	at voluntari		
					9006	1,286 123	# of obset	SUX4 4,11 16	d; ^b Firms tha		
		00-2010			2005	1,427 106		21	umple perio		
	04 to 2010	scal years 200 and ABSDA	tions		PUDG	1,569 102		SUX302 7,367 466	during the se		
	scal years 200 status	ıtabase for fi: calculate DA	ustat sample 1 ten observa		9003	1,576 100		550 X 559 58	ontrol audits		
	abase for fis vitched filing	at Annual da 00-6999) ormation to	t with Comp vith less thar ors		6006	1,509 97	ulations	2, 1	lb internal c d		
	malytics dat irms that sv	he Compust: IC codes 600 sufficient inf	/tics firm lis r-industry w sing regress	us)	1006	1,329 85	<i>l control reg</i> of firms	260 260	opt SOX 404 ample perio		
lection	the Audit A I filers and f	er firms tions from tl ms ompanies (S ns without s	Audit Analy ns from yea ns with mise	3 unique firm	n by year	1,330 73	n by interna # 0	0	t did not add during the s		
Panel A: sample sei	Unique firms from Deleted accelerated	Non-accelerated fil Firm-year observal Deleted non-US fin Deleted financial cc Deleted observation	Compustat sample After merging the . Deleted observatio Deleted observation	Final sample (3,145	Panel B: distributio	Non-adopters ^a Adopters ^b	Panel C. distributio	Non-adopters ^a Adopters ^b	Notes: ^a Firms that one or more years one	Table Sample selection a distributi	I. Ind

accelerated filers. Next, we collect financial data from the Compustat Annual database for the years 2000 to 2010. After excluding 17,056 non-US observations, 23,939 observations from financial companies (SIC codes 6000-6999) and 16,770 observations lacking sufficient information to compute discretionary accruals, the final Compustat sample consists of 48,852 observations. Merger of the non-accelerated filer sample from Audit Analytics with Compustat financial data yields 20,051 observations (4,304 unique firms). We delete 90 observations from year-industry groups with less than 10 observations and 4,603
observations with missing information for the control variables. Our final sample consists of 15,358 observations for 3,148 unique firms. Panel B provides the distribution of the sample by year. Voluntary adoption of Section 404b occurs more frequently during 2006-2010 than in the previous years. Panel C provides the sample distribution across the regimes.

Table II presents descriptive statistics for the full sample in Panel A, and for the subsamples of adopters and non-adopters in Panel B. Adopters have a lower mean (but not median) for ABSDA than non-adopters. Both the mean and median for DA are lower for adopters compared to non-adopters. In addition, the two samples differ on several dimensions. Adopters are larger and are more likely to be audited by the Big N firms than non-adopters. However, non-adopters are financially less stressed compared with the adopters, as indicated by the means for LOSS and the means and medians for CFO. We examined the correlations (untabulated) for the independent variables. None of the correlations exceed 0.5 suggesting that multicollinearity is not a concern[11].

Table III, Panel A presents our regression results for equation (1). In our descriptions relating to this and succeeding tables, we use the labels in Figure 1 to identify the periods we discuss: PRESOX, SOX 302, SOX 404a, SOX 404b and AFTERDROP.

ABSDA (unsigned discretionary accruals) is the dependent variable in Column 1, and DA (signed discretionary accruals) is the dependent variable in Column 2. For the ABSDA regression in Column 1, all seven test indicator variables have significant negative coefficients, indicating lower levels of accruals compared with the PRESOX levels for non-adopters. The DA regression in Column 2 yields similar results, except the insignificant coefficient for ADOPTPRESOX which suggests the level of signed discretionary accruals is not different between the non-adopters and adopters in the PRESOX period.

Table III, Panel B provides tests of incremental benefit of each regime over PRESOX and over the previous regime (RQ1) for the ABSDA analysis. Column 1 indicates that both adopters and non-adopters had lower accruals in the SOX 302 regime relative to the PRESOX period, supporting H1a and suggesting both benefited from SOX 302. Column 4 reports the incremental benefit of SOX 404a over SOX 302. Here, we find that only the non-adopters gained from SOX 404a over SOX 302, providing marginal support for H1b for the non-adopters. The difference in the incremental benefit for the two groups however is not significant. Finally, discretionary accruals is lower in SOX 404b than in SOX 302 for the adopters (-0.365 versus -0.281; p-value < 0.001), supporting H1c.

The effect of SOX 404b can only be seen for the adopters. Column 3 indicates that the adopters of SOX 404b benefited compared to the PRESOX period. The coefficient in Column 5 indicates the incremental benefit of SOX 404b over SOX 404a. This coefficient is significantly negative, supporting *H1d* and indicating that the voluntary adopters of SOX 404b benefited from the internal control audit in the form of higher financial reporting quality. Column 6 reports the differences in the accruals after the voluntary adopters drop the SOX 404 audits. The insignificant coefficient suggests that once the voluntary adopters were audited, the discontinuation of the audit did not make a difference, suggesting that the benefit of the audit continued after such discontinuance[12].



RAF

19,2

Full sample ($V = 15,358$)reporting quality ofVariableMeanMedianSDquality of small firmsDA0.020.010.77DA0.020.010.77LNMVE3.423.501.69BIGN0.470.000.02LITIGATION0.180.000.02LOSS0.601.000.49LACCRUAL-0.19-0.080.88LEVERAGE0.660.490.84MB2.631.549.11CFO-0.090.010.45SDCFO0.230.100.55SDREV0.320.210.36ASALES1.231.041.10Panel B: non-adopters and adoptersNon-adopters ($V = 14,142$)Adopters ($N = 1,216$) I -statistica ^a Non-adopters ($N = 14,142$)Adopters ($N = 1,216$) I -statistica ^a VariableMeanMedianSD(1) (2) vis (6)(1)(2) (3) (4)(5) (6) (7)(8)ABSDA0.190.080.780.160.000.03-3.22***-0.59DA0.030.010.03-3.24***LNMVE3.63.421.714.13LNWVE3.63.421.714.13LNWVE3.63.421.711.000.49DA0.000.030.990.220.23LOSS0.591.000.499	D 1 A. 6 11									Financial
VariableMeanMedianSD.quality of (1) (2) (3) small firms $ABSDA$ 0.19 0.08 0.75 DA 0.02 0.01 0.77 $LNWVE$ 3.42 3.50 1.69 BGN 0.47 0.00 0.50 BGR 0.004 0.00 0.29 $LITIGATION$ 0.18 0.00 0.39 $LOSS$ 0.660 1.00 0.49 $LIACCRUAL$ -0.19 -0.08 0.88 $LEVERAGE$ 0.666 0.49 0.84 $IIACCRUAL$ -0.19 0.01 0.45 $SDCFO$ 0.23 0.10 0.55 $SDREV$ 0.32 0.21 0.36 $ASALES$ 1.23 1.04 1.10 Panel B: non-adopters (N = 14,142)Adopters (N = 1,216) t statistic ⁴ $Vilcoxon.Zb$ $Vilcoxon.Zb$ $Vilcoxon.Zb$ $Variable$ MeanMedianSD $(1) vs (4)$ $(2) vs (5)$ $(1) (2) vs (5)$ $(1) (2) vs (5)$ $(1) 02$ $(3) (4) (5) (6) (7)$ (8) DA 0.03 0.10 0.00 0.03	Panel A: juu san	npie			Б	ull comple	(N - 15)	358)		reporting
ABSDA(1)(2)(3)guallity ofABSDA0.190.080.75DA0.020.010.77LNMVE3.423.501.69BIGN0.470.000.50DA0.0040.000.02LITIGATION0.180.000.39LOSS0.661.000.49LACCRUAL-0.19-0.080.88LEVERAGE0.660.490.84MB2.631.549.11CFO-0.090.010.45SDCFO0.230.210.36ASALES1.231.041.10Panel B: non-adopters (N = 14,142)Adopters (V = 12,16)t-statistic ^a Wilcoxn-Z ^b (1)(2)(3)(4)(5)(6)(7)(1)(2)(3)(4)(5)(6)(7)(1)(2)(3)(4)(5)(6)(7)(1)(2)(3)(4)(5)(6)(7)(1)(2)(3)(4)(5)(6)(7)(3)(4)(5)(6)(7)(8)ASALES1.231.041.10Panel B: non-adopters (N = 14,142)Adopters (V = 14,142)Adopters (V = 12,16)t-statistic ^a (1)(2)(3)(4)(5)(6)(7)(3)(3)0.010.80-0.29-3.289***-0.59DA0.030.01 <td>Variable</td> <td></td> <td>М</td> <td>ean</td> <td>ľ</td> <td>vun sampie N</td> <td>(1) – 15 ledian</td> <td>,556)</td> <td>SD</td> <td>quality of</td>	Variable		М	ean	ľ	vun sampie N	(1) – 15 ledian	,556)	SD	quality of
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	variable			1)		14.	(2)		(3)	quality of
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ABSDA		0.	19			0.08		0.75	small firms
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	DA		0.	02			0.01		0.77	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	LNMVE		3.	42			3.50		1.69	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	BIGN		0.47				0.00		0.50	233
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	MERGER		0.	0004			0.00		0.02	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LITIGATION		0.	18			0.00		0.39	
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	LOSS		0.	60			1.00		0.49	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	L1ACCRUAL		-0.	19		-	-0.08		0.88	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LEVERAGE		0.	66			0.49		0.84	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	MB		2.	63			1.54		9.11	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	CFO		-0.	09			0.01		0.45	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	SDCFO		0.	23			0.10		0.55	
ASALES 1.23 1.04 1.10 Panel B: non-adopters and adopters Non-adopters (N = 14,142) Adopters (N = 1,216) t-statistic ^a Wilcoxon-Z ^b Variable Mean Median SD Mean Median SD (1) vs (4) (2) vs (5) (1) (2) (3) (4) (5) (6) (7) (8) ABSDA 0.19 0.08 0.78 0.16 0.08 0.29 -3.22^{***} -0.59 DA 0.03 0.01 0.80 -0.01 -0.00 0.33 -3.89^{***} -4.626^{****} LNMVE 3.36 3.42 1.71 4.13 4.22 1.32 19.18**** 17.70**** BIGN 0.46 0.00 0.50 0.59 1.00 0.49 9.09*** 8.96*** MERGER 0.0004 0.000 0.02 0.000 0.000 0.22 0.23 LOSS 0.59 1.00 0.49 0.71 1.00 0.45 9.13**** 8.49**** LIACCRUAL -0.19 -0.08 0.91	SDREV		0.	32			0.21		0.36	
Panel B: non-adopters and adoptersNon-adopters (N = 14,142)Adopters (N = 1,216)t-statisticaWilcoxon-ZbVariableMeanMedianSDMeanMedianSD(1) vs (4)(2) vs (5)(1)(2)(3)(4)(5)(6)(7)(8)ABSDA0.190.080.780.160.080.29 -3.22^{***} -0.59 DA0.030.010.80 -0.01 -0.00 0.33 -3.89^{***} -4.626^{***} LNMVE3.363.421.714.134.221.3219.18^{***}17.70^{***}BIGN0.460.000.500.591.000.499.09^{***}8.96^{***}MERGER0.00040.0000.020.0000.0000.00 -2.45^{**} -0.72 LITIGATION0.180.000.390.190.000.390.220.23LOSS0.591.000.490.711.000.459.13^{***}8.49^{***}L1ACCRUAL -0.19 -0.08 0.91 -0.14 -0.09 0.502.85^{***} -2.32^{**} LEVERAGE0.670.490.860.630.490.65 -2.01^{**} 0.13MB2.581.529.043.191.879.852.11**3.97^{***}CFO -0.09 0.010.45 -0.12 -0.01 0.42 -3.04^{***} -5.62^{***} SDCFO0.240.100.57<	ΔSALES		1.	23			1.04		1.10	
Non-adopters (N = 14,142)Adopters (N = 1,216) t -statisticaWilcoxon-Z ^b VariableMeanMedianSDMeanMedianSD(1) vs (4)(2) vs (5)(1)(2)(3)(4)(5)(6)(7)(8)ABSDA0.190.080.780.160.080.29 -3.22^{***} -0.59 DA0.030.010.80 -0.01 -0.00 0.33 -3.89^{***} -4.626^{***} LNMVE3.363.421.714.134.221.3219.18^{***} 17.70^{***} BIGN0.460.000.500.591.000.499.09^{***} 8.96^{***} MERGER0.00040.0000.020.0000.00 -2.45^{**} -0.72 LITIGATION0.180.000.390.120.230.220.23LOSS0.591.000.490.711.000.459.13^{***} 8.49^{***} L1ACCRUAL -0.19 -0.08 0.91 -0.14 -0.09 0.50 2.85^{***} -2.32^{**} LEVERAGE0.670.490.860.630.490.65 -2.01^{**} 0.13MB2.581.529.043.191.879.852.11** 3.97^{***} CFO -0.09 0.010.45 -0.12 -0.01 0.42 -3.04^{***} -5.62^{***} SDCFO0.240.100.570.180.100.29 -5.71^{***} $-6.81^{$	Panel B: non-ad	obters and ac	lobters							
VariableMeanMedianSDMeanMedianSD(1) vs (4)(2) vs (5)(1)(2)(3)(4)(5)(6)(7)(8)ABSDA0.190.080.780.160.080.29 -3.22^{***} -0.59 DA0.030.010.80 -0.01 -0.00 0.33 -3.89^{***} -4.626^{***} LNMVE3.363.421.714.134.221.3219.18^{***} 17.70^{***} BIGN0.460.000.500.591.000.499.09^{***} 8.96^{***} MERGER0.00040.0000.020.0000.000 -2.45^{**} -0.72 LITIGATION0.180.000.390.190.000.390.220.23LOSS0.591.000.490.711.000.459.13^{***} 8.49^{***} L1ACCRUAL -0.19 -0.08 0.91 -0.14 -0.09 0.50 2.85^{***} -2.32^{**} LEVERAGE0.670.490.860.630.490.65 -2.01^{**} 0.13MB2.581.529.043.191.879.852.11**3.97^{***}CFO -0.09 0.010.45 -0.12 -0.01 0.42 -3.04^{***} -5.62^{***} SDCFO0.240.100.570.180.100.29 -5.71^{***} -6.81^{***} SDREV0.330.210.360.260.170.	1 0000 200000 000	Non-ador	pters ($N = 14$.142)	Adopt	ters $(N = 1.2)$	216)	<i>t</i> -statistic ^a	Wilcoxon-Z ^b	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Variable	Mean	Median	SD	Mean	Median	SD	(1) vs (4)	(2) vs (5)	
ABSDA 0.19 0.08 0.78 0.16 0.08 0.29 -3.22^{***} -0.59 DA 0.03 0.01 0.80 -0.01 -0.00 0.33 -3.89^{***} -4.626^{***} LNMVE 3.36 3.42 1.71 4.13 4.22 1.32 19.18^{***} 17.70^{***} BIGN 0.46 0.00 0.50 0.59 1.00 0.49 9.09^{***} 8.96^{***} MERGER 0.0004 0.000 0.02 0.000 0.000 0.00 -2.45^{**} -0.72 LITIGATION 0.18 0.00 0.39 0.19 0.00 0.39 0.22 0.23 LOSS 0.59 1.00 0.49 0.71 1.00 0.45 9.13^{***} 8.49^{***} LIACCRUAL -0.19 -0.08 0.91 -0.14 -0.09 0.50 2.85^{***} -2.32^{**} LEVERAGE 0.67 0.49 0.86 0.63 0.49 0.65 -2.01^{**} 0.13 MB 2.58 1.52 9.04 3.19 1.87 9.85 2.11^{**} 3.97^{***} CFO -0.09 0.01 0.45 -0.12 -0.01 0.42 -3.04^{***} -5.62^{***} SDCFO 0.24 0.10 0.57 0.18 0.10 0.29 -5.71^{***} -6.81^{***} ASALES 1.23 1.05 1.10 1.27 1.04 1.16 1.19 -0.52		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ABSDA	0.19	0.08	0.78	0.16	0.08	0.29	-3.22***	-0.59	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	DA	0.03	0.01	0.80	-0.01	-0.00	0.33	-3.89^{***}	-4.626^{***}	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LNMVE	3.36	3.42	1.71	4.13	4.22	1.32	19.18***	17.70***	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	BIGN	0.46	0.00	0.50	0.59	1.00	0.49	9.09***	8.96***	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	MERGER	0.0004	0.000	0.02	0.000	0.000	0.00	-2.45^{**}	-0.72	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LITIGATION	0.18	0.00	0.39	0.19	0.00	0.39	0.22	0.23	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	LOSS	0.59	1.00	0.49	0.71	1.00	0.45	9.13***	8.49***	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	L1ACCRUAL	-0.19	-0.08	0.91	-0.14	-0.09	0.50	2.85***	-2.32^{**}	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	LEVERAGE	0.67	0.49	0.86	0.63	0.49	0.65	-2.01^{**}	0.13	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	MB	2.58	1.52	9.04	3.19	1.87	9.85	2.11**	3.97***	
SDCFO 0.24 0.10 0.57 0.18 0.10 0.29 -5.71*** -2.12** SDREV 0.33 0.21 0.36 0.26 0.17 0.28 -7.72*** -6.81*** ΔSALES 1.23 1.05 1.10 1.27 1.04 1.16 1.19 -0.52	CFO	-0.09	0.01	0.45	-0.12	-0.01	0.42	-3.04^{***}	-5.62^{***}	
SDREV 0.33 0.21 0.36 0.26 0.17 0.28 -7.72*** -6.81*** ΔSALES 1.23 1.05 1.10 1.27 1.04 1.16 1.19 -0.52	SDCFO	0.24	0.10	0.57	0.18	0.10	0.29	-5.71***	-2.12^{**}	
ASALES 1.23 1.05 1.10 1.27 1.04 1.16 1.19 -0.52	SDREV	0.33	0.21	0.36	0.26	0.17	0.28	-7.72^{***}	-6.81^{***}	
	ASALES	1.23	1.05	1.10	1.27	1.04	1.16	1.19	-0.52	

Notes: Variable definitions are in Appendix. *** (**) [*] Significantly different from zero at or below the 0.01 (0.05) [0.1] level (two-tailed). ^at-statistics for the test of differences between the means of the non-adopters and adopters. ^bZ-statistics for the Wilcoxon Ranksum test between the non-adopters and adopters

 Table II.

 Descriptive statistics

Table III, Panel C presents test results for the DA analysis. The results support *H1a*. Columns 1-3 show that compared to PRESOX, both non-adopters and adopters have lower accruals in SOX 302 and SOX 404a periods, and adopters have lower accruals in SOX 404b period. This suggests that the regulations have improved quality for both groups of firms. Surprisingly, in Column 4 we find that signed discretionary accruals is greater in SOX 404a compared to PRESOX. This is inconsistent with *H1b*. The F-test shows that discretionary accruals is lower in SOX 404b than in SOX 302 (-0.285 versus -0.202; p < 0.01), supporting *H1c*. Consistent with results from the ABSDA analysis, Column 5 indicates that adopters have lower level of discretionary accruals in SOX 404b compared to SOX 404a, supporting H1d and highlighting the incremental benefits brought about by SOX 404b[13].



RAF				
192	Panel A: regression results		Dapanda	nt moriable
15,2			APSDA	nt variable
	Variables		ADSDA (1)	
	CONSTANT	ρ	(1) 0.488*** (0.072)	
	NONADODT202	ρ_0	0.220*** (0.072)	
	NONADOF 1502 NONADOPT404a	ρ_1	-0.320*** (0.003)	
924	ADOPTPPESOY	ρ_2	$-0.301^{++}(0.001)$ 0.146***(0.026)	
234	ADOF IF RESUA	ρ_3	-0.140 (0.030)	
	ADOPT404-	ρ_4	-0.261 (0.004)	
	ADOPT404b	ρ_5	-0.290*** (0.003)	
	ADOF 14040 AFTERDROD	ρ_6	-0.303*** (0.039)	
	AFTERDROF	ρ_7	-0.370^{-11} (0.000)	
	DIGIN		$-0.047^{444}(0.014)$ 0.427*(0.222)	
	MERGER LITICATION		$0.427^{\circ}(0.232)$	
	LITIGATION		$-0.030^{\text{max}}(0.011)$	
	LUSS		-0.026*** (0.013)	
	LIACCRUAL		$-0.042^{*}(0.022)$	
	LEVERAGE		0.020 (0.016)	
	MB		0.001 (0.001)	
	LNMVE		$-0.012^{***}(0.005)$	
	CFO		-0.224^{***} (0.037)	
	SDCFO		0.032 (0.021)	
	SDREV		0.093*** (0.027)	
	ΔSALES		0.021** (0.008)	
	Year FE		Included	
	N		15,358	

Panel B: Test of RQ1: Incremental benefit over PRESOX, and compared with previous regime Benefit (compared with PRESOX) Incremental benefit (compared with previous

	(1	,		regime)	r i i i i i i i i i i i i i i i i i i i
	SOX302	SOX404a	SOX404b	SOX404a	SOX404b	AFTERDROP
	(1)	(2)	(3)	(4)	(5)	(6)
Dependent Variable – ABSDA						
Non-adopters	-0.320 ***	-0.361 ***		-0.041*		
	β_1	β_2		$\beta_2 - \beta_1$		
Adopters	-0.135^{***}	-0.150^{***}	-0.219^{***}	-0.015	-0.069^{**}	-0.005
	$\beta_4 - \beta_3$	$\beta_5 - \beta_3$	$\beta_6 - \beta_3$	$\beta_5 - \beta_4$	$\beta_6 - \beta_5$	$\beta_7 - \beta_6$
Difference (Adopters minus Non-	0.185***	0.211***		0.026		
adopters)	$\beta_4 - \beta_3 - \beta_1$	$\beta_5 - \beta_3 - \beta_2$		$(\beta_{5} - \beta_{4}) -$		
				$(\beta_2 - \beta_1)$		

0.061

Panel C: Test of RQ1: Incremental benefit over PRESOX, and compared with previous regime Benefit (compared with PRESOX) Incremental benefit

				(compare	d with previou	is regime)
	SOX302	SOX404a	SOX404b	SOX404a	SOX404b	AFTERDROP
	(1)	(2)	(3)	(4)	(5)	(6)
Dependent Variable – DA						
Non-adopters	-0.218^{***}	-0.174 ***		0.044**		
	β_1	β_2		$\beta_2 - \beta_1$		
Adopters	-0.148^{***}	-0.108*	-0.231^{***}	0.040	-0.123^{***}	0.02
	$\beta_4 - \beta_3$	$\beta_5 - \beta_3$	$\beta_6 - \beta_3$	$\beta_5 - \beta_4$	$\beta_6 - \beta_5$	$\beta_7 - \beta_6$
Difference (Adopters minus Non-	0.070***	0.066***		-0.04		
adopters)	$\beta_4 - \beta_3 - \beta_1$	$\beta_5 - \beta_3 - \beta_2$		$(\beta_{5} - \beta_{4}) -$		
				$(\beta_2 - \beta_1)$		

Table III. Main analyses

Notes: Variable definitions are in Appendix. Standard errors are in parentheses. *** (**) [*] Significantly different from zero at or below the 0.01 (0.05) [0.1] level (two-tailed)

Adj R-squared

DA (2) 0.363*** (0.074) -0.218*** (0.067) -0.174*** (0.063)

-0.054(0.038) -0.202^{***} (0.068)

-0.162** (0.071) -0.285*** (0.062) -0.265*** (0.063) -0.040 *** (0.014)0.720** (0.314) -0.003 (0.012) -0.126^{***} (0.015) 0.025 (0.024) 0.023 (0.017) 0.001 (0.001) -0.027*** (0.005) -0.117*** (0.043) -0.009(0.022)-0.011(0.027)-0.006 (0.010) Included

15,358

0.018

Table IV addresses *RQ2*. We divide our sample firms into "clean" and ICW firms. We employ a very stringent definition of clean firms, i.e. those that did not have an ICW in any regulation regime. Correspondingly, ICW firms are those that had an ICW in one or more of the years they are in the sample. Panel A, Columns 1 and 2 (Columns 3 and 4) present regression estimates using ABSDA (DA) as the dependent variable. As in Table III Panel A, the coefficients for most of the test variables are negatively significant for both the clean and ICW samples. Panels B and C present, as before, benefits (over the PRESOX regime and incremental benefits of each regime over the previous regime) for the ABSDA and DA analyses, respectively.

Panel B indicates important differences between clean and ICW firms in the benefits from the different regulations. All firms, clean and ICW, adopters and non-adopters, have lower accruals in the SOX 302 period compared with the PRESOX period. Also, consistent with Table III, non-adopters benefited more than adopters for both the clean and ICW samples. All firms have lower accruals in the SOX 404a period relative to PRESOX, and the adopters (both clean and ICW) have lower accruals in the SOX 404b period compared with the PRESOX period. However, the incremental benefits in Columns 4-6 indicate different results for clean and ICW firms. The clean firms (both adopters and non-adopters) show no incremental benefits from SOX 302 to SOX 404a, and the adopters show no incremental benefits from SOX 404a to SOX 404b. For the ICW firms, neither adopters nor non-adopters benefit incrementally from SOX 404a over SOX 302. The only incremental benefit we observe is for the ICW firms that adopted SOX 404b. Accruals for these firms are lower in the SOX 404b period relative to the SOX 404a period. Finally, firms that dropped the SOX 404b audits did not see any change in accruals from the SOX 404b period.

Panel C presents test results using DA as the dependent variable. Consistent with Panel B, it shows that most firms, clean or ICW, have better financial reporting quality in SOX 302, SOX 404a, and SOX 404b compared to PRESOX. Also, consistent with Panel B, ICW firms are the only group that benefited incrementally from SOX 404b. However, Column 4 shows that clean non-adopters have greater accruals moving from SOX 302 to SOX 404a.

Next, we examine whether the results for clean firms differ across firms that began with different reporting quality. The internal control regulations may not have a visible impact on firms that already have good financial reporting quality and have no internal control issues. We partition the clean firms into quintiles based on their average ABSDA in the PRESOX period. We then estimate our model for each quintile and include our results in Table V Panel A. Columns 1-3 present ABSDA analysis regression estimates for the 1st, 3rd and 5th quintiles, and Columns 4-6 present the DA analysis results.

In the ABSDA regression estimates, NONADOPT302 and ADOPT302 have significant negative coefficients for the 1st quintile, indicating that both non-adopters and adopters have lower levels of accruals in the SOX 302 period compared with non-adopters in the PRESOX period. This evidence is consistent with those provided in Cohen *et al.* (2008) that the quality of accruals was lower for the population of firms in the PRESOX period. However, unlike in Tables III and IV, the estimated coefficients for NONADOPT404a and AFTERDROP in the 1st quintile are positive and significant, indicating higher levels of accruals compared with the PRESOX levels for non-adopters. None of the test indicator variables have significant coefficients for the 3rd quintile, implying similar levels of accruals compared with the PRESOX levels for non-adopters. Quintile 5 is the only quintile with negative significant coefficients for all seven test indicator variables. For the DA analysis, firms in the 1st and 5th quintiles have lower accruals in the SOX 302 period compared to the PRESOX period. In the 3rd quintile, adopters are the only group with lower accruals in the SOX 302 period compared to the PRESOX period. Similar to the ABSDA analysis, the



Financial reporting quality of small firms

RAF								
10.9					Dependent	Variable		
19,2				ABS	DA	~	D	A
	X7 · 11		Clean fi	rms	ICW firms	Clean	firms	ICW firms
	Variables	ulta	(1)		(2)	(3)	(4)
	CONSTANT	uus	<i>B</i> 0.414***	(0.073)	0.673*** (0.170	0 303**	* (0.075)	0 549*** (0 184)
	NONADOPT302		$\rho_0 = 0.414$ $\beta_1 = 0.975***$	(0.075)	-0.468*** (0.173) _0.303	* (0.064)	_0.357** (0.181)
236	NONADOPT404a		$\beta_1 = 0.275$ $\beta_2 = -0.361***$	(0.056)	-0.519*** (0.170	-0.113	* (0.058)	-0.352** (0.178)
	ADOPTPRESOX		$\beta_2 = 0.001$ $\beta_3 = -0.131^{***}$	(0.037)	-0.179** (0.087	-0.05	4 (0.038)	-0.064 (0.093)
	ADOPT302		$\beta_4 = -0.249^{***}$	(0.064)	-0.375** (0.171) -0.194**	* (0.064)	-0.243(0.181)
	ADOPT404a		$\beta_5 -0.256^{***}$	(0.064)	-0.425** (0.168) -0.175*	* (0.069)	-0.392** (0.190)
	ADOPT404b		$\beta_6 -0.312^{***}$	(0.056)	-0.516*** (0.162) -0.139*	* (0.058)	-0.506*** (0.168)
	AFTERDROP		$\beta_7 = -0.312^{***}$	(0.057)	-0.549*** (0.170) -0.156**	* (0.060)	-0.450** (0.177)
	BIGN		-0.036^{***}	(0.012)	-0.073* (0.039) -0.027*	* (0.013)	-0.080** (0.041)
	MERGER		0.574**	(0.271)	-0.120^{**} (0.061) -0.833*	* (0.356)	-0.157** (0.068)
	LITIGATION		-0.044^{***}	(0.011)	-0.005 (0.026) -0.00	9 (0.012)	0.002 (0.029)
	LOSS		-0.001	(0.014)	-0.061** (0.026) -0.118**	* (0.015)	-0.140*** (0.029)
	LIACCRUAL		-0.027	(0.019)	-0.055 (0.038	6) 0.00	7 (0.021)	0.039 (0.042)
	LEVERAGE		0.010	(0.009)	0.036 (0.040) 0.00	3 (0.011)	0.053 (0.041)
	MB		0.0003	(0.001)	0.002 (0.001) 0.00	1 (0.001)	0.001 (0.002)
	CEO		-0.012***	(0.006)	-0.014 (0.012	-0.024^{**}	* (0.006) * (0.041)	-0.035*** (0.013)
	SDCFO		-0.113***	(0.033)	0.001 (0.029	0.009	5 (0.030)	-0.138* (0.033)
	SDREV		0.050	(0.030)	0.062 (0.040	0.01	7 (0.030)	-0.034 (0.032)
	ASALES		0.032***	(0.007)	-0.002 (0.019	-0.00	2 (0 009)	0.043 (0.042)
	Year FE		Included	1	Included	Include	2 (0.000) ed	Included
	N		10.456	5	4.902	10.45	56	4.902
	Adj R-squared		0.052	2	0.085	0.01	17	0.026
	Panel B: Test of RQ2: In	cremental benefit (comp SOX302	<i>it over PRESOX</i> Benefit pared with PRE SOX404a	and com SOX) SOX4	npared with previo (c 404b SOX4	ous regime Increme compared wit	ental bene h previou SOX404b	efit 1s regime) AFTERDROP
		(1)	(2)	(3)) (4)	(5)	(6)
	Dependent Variable – A	BSDA						
	Non adopters	0 275***	0.307***		0.0	29		
	1011-adopters	B1	B2		β ₂ -	β ₁		
	Adopters	-0.118**	-0.125**	-0.18	1*** -0.0	007	-0.056	0.000
	*	$\beta_4 - \beta_3$	$\beta_5 - \beta_3$	$\beta_6 -$	$\beta_3 \qquad \beta_5 -$	β_4	$\beta_6 - \beta_5$	$\beta_7 - \beta_6$
	Difference	0.157***	0.182***		0.02	25		
	(Adopters minus Non-	$\beta_4 - \beta_3 - \beta_1$	$\beta_5 - \beta_3 - \beta_2$		$\beta_5 - \beta_4 -$	$\beta_2 - \beta_1$		
	adopters)							
	ICW Firms							
	Non-adopters	-0.468***	-0.519***		-0.0)51		
	A 1 4	β_1	β_2	0.00	$\beta_2 - \beta_2 $	β_1	0.001**	0.022
	Adopters	-0.196*	-0.240	-0.33	-0.	00 -	-0.091	-0.033
	Difference	$p_4 - p_3$ 0.979***	P 5-P 3 0 272***	μ_{6}	-μ ₃ μ ₅ -	p_4	$p_{6} - p_{5}$	$p_7 - p_6$
	(Adopters minus Non	$B_1 = R_2 = R$	BR_R		BR	$\beta_{0} = \beta_{1}$		
Table IV.	adopters)	$\rho_4 \ \rho_3 \ \rho_1$	$P_5 P_3 P_2$		$P_5 P_4$	$P_2 P_1$		
Clean firms and ICW	adopteroj							(the th
firms								(continued)



		Benefit		Incre	Incremental benefit				
	(compared with PRESOX)		OX)	(compared	with previous	regime)	small firm		
	SOX302	SOX404a	SOX404b	SOX404a	SOX404b	AFTERDROP			
	(1)	(2)	(3)	(4)	(5)	(6)			
Dependent Variable – L	DA						0.07		
Clean Firms							237		
Non-adopters	-0.171^{***}	-0.113*		0.058**					
	β_1	β_2		$\beta_2 - \beta_1$					
Adopters	-0.14^{***}	-0.121*	-0.085*	0.019	0.036	-0.017			
	$\beta_4 - \beta_3$	$\beta_5 - \beta_3$	$\beta_6 - \beta_3$	$\beta_5 - \beta_4$	$\beta_6 - \beta_5$	$\beta_7 - \beta_6$			
Difference	0.031***	-0.008		-0.039					
(Adopters minus Non-	$\beta_4 - \beta_3 - \beta_1$	$\beta_5 - \beta_3 - \beta_2$		$\beta_{5} - \beta_{4} - \beta_{2} - \beta_{1}$					
adopters)									
ICW Firms									
Non-adopters	-0.357 ***	-0.352^{***}		0.005					
	β_1	β_2		$\beta_2 - \beta_1$					
Adopters	-0.179	-0.328 **	-0.442^{***}	-0.149	-0.114^{***}	0.056			
	$\beta_4 - \beta_3$	$\beta_5 - \beta_3$	$\beta_6 - \beta_3$	$\beta_5 - \beta_4$	$\beta_6 - \beta_5$	$\beta_7 - \beta_6$			
Difference	0.178***	0.024**		-0.154					
(Adopters minus Non-	$\beta_4 - \beta_3 - \beta_1$	$\beta_5 - \beta_3 - \beta_2$		$\beta_{5} - \beta_{4} - \beta_{2} - \beta_{1}$					
adopters)									

different from zero at or below the 0.01 (0.05) [0.1] level (two-tailed)

Table IV.

Financial

coefficient on ADOPT404b is negative and significant only for Quintile 5, firms with the worst financial reporting quality in the PRESOX period.

In Panel B of Table V, the ABSDA analysis shows that there are significant variations across the quintiles in the benefits from the different regulation regimes. Quintile 1 comprises firms with the lowest accruals in the PRESOX period. All firms in the 1st quintile, adopters and non-adopters, have lower accruals in the SOX 302 period compared with the PRESOX period, and the change in accruals is similar between the adopters and non-adopters. Columns 2 and 4 exhibit higher accruals in the SOX 404a period relative to both the PRESOX and SOX 302 periods for the non-adopters, indicating (unexpectedly) deterioration in quality for the non-adopters in the SOX 404a periods, but have similar accruals in the PRESOX and SOX 404a periods (possibly because the incremental gains from SOX 302 are offset by the reduction in quality during the SOX 404a period). Lastly, adopters that dropped SOX 404b audits show no change in accruals from the SOX 404b period.

In Quintile 3, which includes firms with "medium" performance in the PRESOX period, non-adopters have lower accruals in the SOX 302 period relative to the PRESOX period, while adopters have similar levels of accruals in the two periods. Column 2 indicates that non-adopters have lower accruals in the SOX 404a period compared with the PRESOX period, but adopters have significant higher accruals in the SOX 404a period compared with the PRESOX period. Neither non-adopters nor adopters show any incremental benefits from SOX 404a or SOX 404b. Column 6 shows that there is no change in accruals for firms that dropped 404b audits.

Quintile 5 comprises firms with the highest accruals in the PRESOX period. All firms in Quintile 5 have lower accruals in the SOX 302 and SOX 404a periods compared with the



RAF 19,2 238	trie 5th Quintile (6) (035) (1362**** (0.426) (035) (1362**** (0.426) (022) $-0.594*** (0.296)$ (075) $-0.534** (0.296)$ (075) $-0.333 (0.322)$ (075) $-0.333 (0.323)$ (050) $-0.919** (0.298)$ (030) $-0.919** (0.298)$ (031) $-0.785** (0.342)$ d Included 1,467 0.046 heft nefit nefit AFTERDROP (6) (6) (6) (6) (7) -0.056 5 $\beta_{7}-\beta_{6}$	Continued
	riable DA ile DA (5) (026) (133***(0.0133**() 0.012 (0 0.012 (0 0.013 (0 0.	$2-\beta_1$
	$\begin{array}{c} \mbox{Dependent Va} \\ Dependent$	$eta_5 - eta_4 - eta_2$
	$\begin{array}{ccccc} & 5 th Quintile \\ & 5 th Quintile \\ & (3) \\ & (3) \\ & (4) \\ & (6) \\ & (-1.607^{***}, (0.2) \\ & (1.552^{***}, (0.2) \\ & (1.538^{***}, (0.2) \\ & (1.467 \\ & (1.467 \\ & (1.467 \\ & (1.467 \\ & (1.467 \\ & (1.467 \\ & (1.344 \\ & (1.467 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & (1.346 \\ & (1.344 \\ & $	5
	ABSDA ABSDA ABSDA ABSDA 229 229 229 229 229 229 229 22	$\beta_{5}-\beta_{3}-\beta_{3}$
	$ \begin{array}{c} \label{eq:constraint} & \label{eq:constraint} \\ \begin{tabular}{c} & \bedin{tabular}{$	$\beta_4 - \beta_3 - \beta$
Table V. Clean firms – partition by average PRESOX quality	Panel A: Regression Results Variables Constant nonadopt404a adopt404a adopt404a adopt404b adopt404b AFTERdrop Control variables Year FE N Adip R-squared Panel B: Test of RQ1: Incremental benefit Non-adopters Adopters Dependent Variable: ABSDA Ist Quintile Non-adopters Adopters minus Non-adopters) - 3rd Quintile Non-adopters Adopters Difference Difference	(Adopters minus Non-adopters)
Clean firms – partition by average PRESOX quality	Panel A: Regr. Panel A: Regr. Variables Constant nonadopt302 adopt404a adopt404b AFTERdrop Control variat. N Adiptescurec Panel B: Test. Non-adopters Non-adopters Panel B: Test. Adopters Difference Adopters Onitifie Non-adopters Adopters Difference Adopters Difference Difference	(Adopters mir

WWW.

are in parentheses. **** (***) [**] Significantly dimerent irom zero at or below tue ovor ty.	$\beta_1 \qquad \begin{array}{c} 0.551^{***} \\ \beta_5 - \beta_3 - \beta_2 \\ \vdots \\ $	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\beta_{1} \qquad \beta_{5} - \beta_{3} - \beta_{2} \qquad \beta_{5} - \beta_{4} - \beta_{2} - \beta_{1} \\ -0.681^{**} \qquad -0.087$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\beta_1 = \beta_5 - \beta_2 - \beta_2 = \beta_5 - \beta_4 - \beta_2 - \beta_1 - 0.019 = 0.040$	$egin{array}{cccccccccccccccccccccccccccccccccccc$	$\beta_2 = -0.006 = -0.008 = 0.133 = -0.008 = -0.00$	* -0.006 0.161***	Benefit Increment (compared with PRESOX) (compared with 1 404a 404b 404a (2) (3) (4)	$\beta_1 \qquad \begin{array}{c} 0.507^{***} & -0.015 \\ \beta_5 - \beta_3 - \beta_2 & \beta_5 - \beta_4 - \beta_2 - \beta_1 \\ & & & & & \\ \end{array}$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$
NOLES: Variable definitions are in Appendix, otandar u ci tots,	Difference 0.517**** 0.517**** 0.517**** 0.517**** 0.517**** 0.517**** 0.517**** 0.517****	Adopters $\beta_1 = -0.077$ $\beta_{-1-\beta_3} = -0.077$ $\beta_{-1-\beta_3} = \beta_{-1-\beta_3} = \beta_{-1-\beta_3$	Adopters minus Non-adopters) $\beta_{4} - \beta_{3} - \beta_{1} \beta_{1} Quintite -0.594^{***}$	dopters -0.139^{μ} $\beta_4 - \beta_3$ fifterence -0.080	(dopters minus Non-adopters) $\beta_{4}-\beta_{3}-\beta_{3}-\beta_{4}$ d Quintile -0.059 on adopters	$\beta_4 - \beta_3$ (fference 0.028^{***}	β_1 –0.139**	ependent Variable – DA it Quintile –0.167**** on adopters –0.167****	302 (1)	ifference 0.522^{***} dopters minus Non-adopters) $\beta_{4} - \beta_{3} - \beta_{3}$	on adopters -1.552^{****} β_1 dopters $\beta_{-1.030}^{****}$

PRESOX period. Non-adopters benefited more than adopters during both the SOX 302 and SOX 404a periods. Column 3 shows that adopters have lower accruals in the SOX 404b period relative to the PRESOX period. Columns 4-6 indicate different results from the previous quintiles. Adopters show lower accruals in the SOX 404b period compared with the SOX 404a period. Quintile 5 is the only group where firms show incremental benefits from SOX 404b over SOX 404a. Finally, firms that dropped SOX 404b audit do not show any changes in accruals, suggesting that the audits may have put sufficient mechanisms in place to retain the benefits of improved internal control quality in future years[14].

Table V, Panel C presents test results of the DA analysis. It shares several common messages with the ABSDA analysis results. First, most firms benefited incrementally from SOX 302 as shown by the negative and significant coefficients in Column 1. Second, quality unexpectedly deteriorated from SOX 302 to SOX 404a period for non-adopters starting with the highest quality in the 1st quintile. Lastly, firms in Quintile 5 are the only group that show incremental benefits moving from SOX 404a to SOX 404b, and the benefits remain even after SOX 404b audits are dropped.

5. Conclusions

RAF

19.2

240

A controversial provision of the Sarbanes–Oxley Act is the requirement relating to the audits of internal control under Section 404b. In 2010, the Dodd–Frank Act permanently exempted non-accelerated filers from complying with Section 404b. The exemption is largely driven by concerns about the substantial compliance costs for small companies. Among those that opposed the permanent exemption include the Center for Audit Quality and the Council of Institutional Investors, maintaining that the exemption "would substantially impact the quality and timing of financial disclosures by sizable public companies to the detriment of investors and our capital markets more generally" [Center for Audit Quality and Council of Institutional Investors (CAQ-CII), 2014]. The Dodd–Frank Act also required the U.S. Government Accountability Office (GAO) to examine the "impact of the permanent exemption on the quality of financial reporting by small public companies and on investors." The GAO subsequently issued a report recommending that the "SEC consider requiring public companies, where applicable, to explicitly disclose whether they obtained an auditor attestation of their internal controls" (GAO, 2013).

In this study we investigate the impact of a series of internal control regulations introduced by SOX (SOX 302, SOX 404a and SOX 404b) on the financial reporting quality of small firms. The issue is important because the incidence of fraud is high among small companies (Beasley *et al.*, 1999) and small companies are likely to have weak internal controls (Doyle *et al.*, 2007a)[15].

We document the following findings. In general, the non-accelerated filers benefited from SOX 302 and SOX 404a regulations over the PRESOX period. This is consistent with evidence provided in Kinney and Shepardson (2011) and other studies. However, we also find that firms that voluntarily adopted SOX 404b, particularly those with material weaknesses, and those without material weaknesses that had low reporting quality, benefited from the internal control audits. Our findings can inform the current policy debate on 404b exemption.

Notes

- More recently, the SEC (SEC, 2019) has issued a proposal for comments to exempt additional firms from SOX 404b audits. Two recent articles also highlight adverse effects of exempting small companies from internal control audits (Levitt, 2019; Michaels, 2019).
- 2. Using ICW disclosure rates as a measure of quality can be problematic. Rice and Weber (2012) document that "a large proportion of firms with material weaknesses fail to report those



weaknesses in a timely manner" with many revealing material weaknesses only after a restatement has been announced. McVay (2011) notes in her discussion of Kinney and Shepardson's (2011) paper, "it is likely that the material weakness disclosure rates under 404(a) are overstated"[...] possibly because "many managers of small firms were lobbying to have Section 404(b) permanently rescinded, they likely exercised additional care and effort when making their assessments in order to convince regulators that Section 404(b) was superfluous."

- 3. In 2012, the House of Representatives introduced a Bill proposing that companies with market capitalization of up to \$250m be exempted from the internal control audits requirement [American Institute of Certified Public Accountants (AICPA), 2012]. This bill was passed by the House Financial Services Committee (Harkins, 2014). More recently, the Financial Choice Act of 2017 includes a proposal to exempt companies with market capitalization less than \$500m from internal control audits and to grant temporary exemption to companies that are not large accelerated filers and whose average annual gross revenues is less than \$50m. The Center for Audit Quality, the Council of Institutional Investors, and the CFA Institute oppose this bill urging "Congress to recognize that capital formation is impacted by many factors and to preserve Section 404(b) on behalf of all investors. Shareholders in public companies of all sizes should benefit from the same protections, including the beneficial, third-party inspection of a company's ICFR that is mandated under SOX Section 404(b)" [Center for Audit Quality, Council of Institutional Investors, 2017].
- 4. A company's "public float" is the aggregate market value of voting and non-voting common equity held by non-affiliates. Accelerated filers are companies with public float of at least \$75m that have been subject to Exchange Act reporting requirements for at least 12 calendar months, have filed at least one annual report, and are not eligible to use Forms 10-KSB and 10-QSB for their annual and quarterly reports.
- 5. Some studies find that small accelerated filers pay disproportionately more to comply with Section 404b than large firms (Krishnan *et al.*, 2008; SEC, 2009).
- 6. See Schneider et al. (2009) for a comprehensive review of studies relating to internal control.
- 7. The SEC's final rule states "[...] we are not requiring any particular procedures for conducting the required review and evaluation. Instead, we expect each issuer to develop a process that is consistent with its business and internal management and supervisory practices."
- 8. Critics of SOX's provisions argue that "the law forces all companies to undergo a financial root canal because a handful had rotten teeth" while others have described "it as 'chemotherapy' to prevent the cancer from recurring after cutting out corporate tumors at Enron, WorldCom and elsewhere. Chemotherapy has unpleasant side effects, but it beats the alternative [...]" (Jones, 2003).
- 9. As a robustness check, we re-run our tests using performance-matched discretionary accruals (Kothari *et al.*, 2005). We first estimate the residual from equation (2). Next, we match each firmyear observation with another from the same two-digit SIC and year with the closest ROA. ABSDA is then calculated as the absolute value of the difference between the residual and the matched firm's residual. Our results and inferences remain similar using this alternative specification.
- 10. We winsorize continuous independent variables at the 1 and 99 per cent levels.
- 11. The highest variance inflation factor (except for some year indicator variables) is 1.45, which is well below the acceptable threshold of 10.
- 12. We conducted additional analysis to examine whether the benefit remains post 2010 (the Dodd– Frank Act permanently exempted non-accelerated filers from 404b audits in 2010). We extended the sample period to 2012, the year the JOBS Act provided a five-year 404b exemption to more emerging growth companies. We included an indicator variable AFTER10, which equals one for



Financial reporting quality of small firms

RAF 19,2	adopter observations after 2010. The coefficient on AFTER10 is negative and significant. It is marginally less negative than AFTERDROP, suggesting decreased financial reporting quality for the adopters after 2010.
	13. We also performed an additional check by including only one indicator variable for the SOX 302

- period (coded 1 for NONADOPT302 or ADOPT302) and one for the SOX 404a period (coded 1 for NONADOPT404a or ADOPT404a). Estimated coefficients on all the test variables are negative and significant at the 1 per cent level, consistent with our main analyses.
- 14. As a robustness test, we also partition the firms based on their ABSDA when they first entered the sample. The results remain qualitatively similar to those tabulated. In a separate robustness test, we divide the sample into quartiles and re-run the analysis. The results are similar to the main analysis using quintiles. Adopters with the lowest reporting quality is the only group with incremental benefits moving to SOX 404b.
- 15. For example, a recent \$31m fraud that occurred in Koss Corporation is directly attributable to weak internal controls even though the management had certified to its effectiveness. Koss was not subject to 404b audit at that time (http://ww2.cfo.com/accounting-tax/2010/01/fraud-casefeeds-sarbox-exemption-critics/).

References

- Alexander, C.R., Bauguess, S.E., Bernile, G., Lee, Y.-H.A. and Marietta-Westberg, J. (2013), "Economic effects of SOX section 404 compliance: a corporate insider perspective", *Journal of Accounting* and Economics, Vol. 56 Nos 2/3, pp. 267-290.
- American Institute of Certified Public Accountants (AICPA) (2012), "News highlights for October 2012", Journal of Accountancy, available at: www.journalofaccountancy.com/Issues/2012/Oct/ Highlights.htm
- Ashbaugh, H., LaFond, R. and Mayhew, B.W. (2003), "Do nonaudit services compromise auditor independence? Further evidence", *The Accounting Review*, Vol. 78 No. 3, pp. 611-639.
- Ashbaugh-Skaife, H., Collins, D.W., Kinney, W.R., Jr and LaFond, R. (2008), "The effect of SOX internal control deficiencies, and their remediation on accrual quality", *The Accounting Review*, Vol. 83 No. 1, pp. 217-250.
- Ashbaugh-Skaife, H., Collins, D.W., Kinney, W.R., Jr and LaFond, R. (2009), "The effect of SOX internal control deficiencies on firm risk and cost of equity", *Journal of Accounting Research*, Vol. 47 No. 1, pp. 1-43.
- Beasley, M.S., Carcello, J.V. and Hermanson, D.R. (1999), Fraudulent Financial Reporting: 1987-1997, an Analysis of U.S. public Companies, COSO, New York, NY.
- Beneish, M.D., Billings, M.B. and Hodder, L.D. (2008), "Internal control weaknesses and information uncertainty", *The Accounting Review*, Vol. 83 No. 3, pp. 665-703.
- Cassell, C.A. Myers, L.A. and Zhou, J. (2013), "The effect of voluntary internal control audits on the cost of capital", Working paper, University of Arkansas and University of Hawaii at Manoa, available at SSRN: http://ssrn.com/abstract=1734300
- Center for Audit Quality and Council of Institutional Investors (CAQ-CII) (2014), "Letter submitted to the house financial services committee", available at: www.cii.org/files/issues_and_advocacy/ correspondence/2014/05_06_14_CAQ_CII_letter_to_HFSC.pdf
- Center for Audit Quality, Council of Institutional Investors, and CFA Institute (CAQ-CII-CFA) (2017), "Letter submitted to the house financial services committee", available at: www.thecaq.org/caqcii-and-cfa-institute-submit-joint-letter-financial-choice-act
- Chen, L.-H., Krishnan, J., Sami, H. and Zhou, H. (2013), "Auditor attestation under SOX section 404 and earnings informativeness", *Auditing: A Journal of Practice and Theory*, Vol. 32 No. 1, pp. 61-84.



- Clayton, J. (2019), Statement at Open Meeting on Proposed Amendments to Sarbanes Oxley 404(b) Accelerated Filer Definition, SEC Washington, DC, available at: www.sec.gov/news/publicstatement/statement-clayton-050919
- Cohen, D.A., Dey, A. and Lys, T.Z. (2008), "Real and accrual-based earnings management in the preand post-Sarbanes–Oxley periods", *The Accounting Review*, Vol. 83 No. 3, pp. 757-787.
- Costello, A.M. and Wittenberg-Moerman, R. (2011), "The impact of financial reporting quality on debt contracting: evidence from internal control weakness reports", *Journal of Accounting Research*, Vol. 49 No. 1, pp. 97-136.
- Dhaliwal, D., Hogan, C., Trezevant, R. and Wilkins, M. (2011), "Internal control disclosures, monitoring, and the cost of debt", *The Accounting Review*, Vol. 86 No. 4, pp. 1131-1156.
- Donaldson, W.H. (2005), "Testimony concerning the impact of the Sarbanes–Oxley act", *available at*: www.sec.gov/news/testimony/ts042105whd.htm
- Doogar, R., Sivadasan, P. and Solomon, I. (2010), "The regulation of public company auditing: evidence from the transition to AS5", *Journal of Accounting Research*, Vol. 48 No. 4, pp. 795-814.
- Doyle, J., Ge, W. and McVay, S. (2007a), "Determinants of weaknesses in internal control over financial reporting", *Journal of Accounting and Economics*, Vol. 44 Nos. 1/2, pp. 193-223.
- Doyle, J., Ge, W. and McVay, S. (2007b), "Accruals quality and internal control over financial reporting", *The Accounting Review*, Vol. 82 No. 5, pp. 1141-1170.
- GAO (2013), SEC Should Consider Requiring Companies to Disclose Whether They Obtained an Auditor Attestation, GAO, Washington, DC, available at: www.gao.gov/assets/660/655710.pdf
- Goh, B.W. and Li, D. (2011), "Internal controls and conditional conservatism", *The Accounting Review*, Vol. 86 No. 3, pp. 975-1005.
- Gordon, L.A. and Wilford, A.L. (2012), "An analysis of multiple consecutive years of material weaknesses in internal control", *The Accounting Review*, Vol. 87 No. 6, pp. 2027-2060.
- Hammersley, J.S., Myers, L.A. and Shakespeare, C. (2008), "Market reactions to disclosure of internal control weaknesses and to the characteristics of those weaknesses under section 302 of the Sarbanes–Oxley act of 2002", *Review of Accounting Studies*, Vol. 14 Nos 2/3, pp. 401-439.
- Harkins, J. (2014), "Fitzpatrick's fostering innovation act passes committee, (June 16)", available at: www.politicspa.com/fitzpatricks-fostering-innovation-act-passes-committee-video/58772/
- Hermanson, D.R. and Ye, Z. (2009), "Why do some accelerated filers with SOX section 404 material weaknesses provide early warning under section 302?", *Auditing: A Journal of Practice and Theory*, Vol. 28 No. 2, pp. 247-271.
- Hoitash, U., Hoitash, R. and Bedard, J.C. (2009), "Corporate governance and internal control over financial reporting: a comparison of regulatory regimes", *The Accounting Review*, Vol. 84 No. 3, pp. 839-867.
- Holder, A.D., Karim, K.E. and Robin, A. (2013), "Was Dodd–Frank justified in exempting small firms from section 404b compliance?", Accounting Horizons, Vol. 27 No. 1, pp. 1-22.
- Hribar, P. and Nichols, D.C. (2007), "The use of unsigned earnings quality measures in tests of earnings management", *Journal of Accounting Research*, Vol. 45 No. 5, pp. 1017-1053.
- Jackson, R.J. Jr (2019), Statement on Proposed Amendments to Sarbanes Oxley 404(b) Accelerated Filer Definition, SEC, Washington, DC, available at: www.sec.gov/news/public-statement/jacksonstatement-proposed-amendments-accelerated-filer-definition
- Jones, J.J. (1991), "Earnings management during import relief investigations", Journal of Accounting Research, Vol. 29 No. 2, pp. 193-228.
- Jones, D. (2003), "Sarbanes–Oxley: dragon or white knight?", USA Today, available at: www.usatoday. com/money/companies/regulation/2003-10-19-sarbanes_x.htm
- Kim, J., Song, B.Y. and Zhang, L. (2011), "Internal control weakness and bank loan contracting: evidence from SOX section 404 disclosures", *The Accounting Review*, Vol. 86 No. 4, pp. 1157-1188.



RAF 19,2	Kinney, W.R., Jr and Shepardson, M.L. (2011), "Do control effectiveness disclosure require SOX 404(b) internal control audits? A natural experiment with small public companies", <i>Journal of Accounting Research</i> , Vol. 49 No. 2, pp. 413-448.							
	Kothari, S.P., Leone, A.J. and Wasley, C.E. (2005), "Performance matched discretionary accrual measures", <i>Journal of Accounting and Economics</i> , Vol. 39 No. 1, pp. 163-197.							
244	Krishnan, G.V. and Yu, V. (2012), "Do small firms benefit from auditor attestation of internal control effectiveness?", <i>Auditing: A Journal of Practice and Theory</i> , Vol. 34 No. 1, pp. 115-137.							
	Krishnan, J., Rama, D. and Zhang, Y. (2008), "Cost to comply with SOX section 404", <i>Auditing: A Journal of Practice and Theory</i> , Vol. 27 No. 1, pp. 169-186.							
	Levitt, A. (2019), "The SEC threatens to go soft on internal controls: a proposal to exempt certain companies from a full audit would encourage fraud and hurt investors", <i>The Wall Street Journal</i> , available at: www.wsj.com/articles/the-sec-threatens-to-go-soft-on-internal-controls-11566158647							
	McVay, S.E. (2011), "Discussion of do control effectiveness disclosures require SOX 404(b) internal control audits? A natural experiment with small U.S. Public companies", <i>Journal of Accounting Research</i> , Vol. 49 No. 2, pp. 449-456.							
	Michaels, D. (2019), "SEC plan gives audit relief to firms that wiped out over \$290 million: proposed exemption has drawn criticism from accountants but praise from businesses", <i>The Wall Street</i> <i>Journal</i> , available at: www.wsj.com/articles/sec-plan-gives-audit-relief-to-firms-that-wiped-out- over-290-million-11564141784							
	Nagy, A.L. (2010), "Section 404 compliance and financial reporting quality", Accounting Horizon Vol. 24 No. 3, pp. 441-454.							
	Ogneva, M., Subramanyam, K.R. and Raghunandan, K. (2007), "Internal control weakness and cost of equity: evidence from SOX section 404 disclosures", <i>The Accounting Review</i> , Vol. 82 No. 5, pp. 1255-1297.							
	Public Company Accounting Oversight Board (PCAOB) (2004), Auditing Standard No. 2: An Audit of Internal Control over Financial Reporting Performed in Conjunction with an Audit of Financial Statements, PCAOB, Washington, DC, available at: https://pcaobus.org/Standards/Archived/ Pages/Auditing_Standard_2.aspx							
	Public Company Accounting Oversight Board (PCAOB) (2006), An Audit of Internal Control over Financial Reporting That is Integrated with an Audit of Financial Statements and Related Other Proposals, PCAOB, Washington, DC, available at: http://pcaobus.org/Rules/Rulemaking/Pages/ Docket021.aspx							
	Rice, S.C. and Weber, D.P. (2012), "How effective is internal control reporting under SOX 404? Determinants of the (non-)disclosure of existing material weaknesses", <i>Journal of Accounting Research</i> , Vol. 50 No. 3, pp. 811-843.							
	Roisman, E.L. (2019), <i>Statement at Open Meeting on Proposed Amendments to Sarbanes Oxley 404(b)</i> <i>Accelerated Filer Definition</i> , SEC, Washington, DC, available at: www.sec.gov/news/public- statement/roisman-statement-proposed-amendments-sox-404b-accelerated-filer-definition							
	Schneider, A., Gramling, A.A., Hermanson, D.R. and Ye, Z. (2009), "A review of academic literature on internal control reporting", <i>Journal of Accounting Literature</i> , Vol. 28, pp. 1-46.							
	SEC (2005), <i>Revisions to Accelerated Filer Definition and Accelerated Deadlines for Filing Periodic Reports</i> , SEC, Washington, DC, available at: www.sec.gov/rules/final/33-8644.pdf							
	SEC (2009), Study of the Sarbanes–Oxley Act of 2002 Section 404 Internal Control over Financial Reporting Requirements. Office of Economic Analysis, SEC, Washington, DC, available at: www. sec.gov/news/studies/2009/sox-404_study.pdf							
	SEC (2011), Study and Recommendations on Section 404(b) of the Sarbanes–Oxley Act of 2002 for Issuers with Public Float between \$75 and \$250 Million, SEC, Washington, DC, available at: www.sec.gov/news/studies/2011/404bfloat-study.pdf							
للاستشارات	المنارخ							

SEC (2019), Amendments to the Accelerated Filer and Large Accelerated Filer Definitions, US Securities and Exchange Commission, Washington, DC, available at: www.sec.gov/rules/proposed/2019/ 34-85814.pdf	Financial reporting
Securities and Exchange Commission (SEC) (2003), <i>Final Rule: Management's Report on Internal Control over Financial Reporting and Certification of Disclosure in Exchange Act Periodic Reports</i> , SEC, Washington, DC, available at: www.sec.gov/rules/final/33-8238.htm	small firms
Teoh, S., Welch, I. and Wong, T.J. (1998), "Earnings management and the underperformance of seasoned equity offerings," <i>Journal of Financial Economics</i> , Vol. 50 No. 1, pp. 63-99	245

Corresponding author Jagan Krishnan can be contacted at: krish@temple.edu



RAF 19,2	Appendix	
246	DA ABSDA NONADOPT302 NONADOPT404a	Discretionary accruals adjusted for portfolio performance Absolute value of discretionary accruals adjusted for portfolio performance Coded 1 for non-adopter observations during years 2002 to 2006, 0 otherwise Coded 1 for non-adopter observations for 2007 and later years, 0 otherwise
	ADOPTPRESOX ADOPT302	Coded 1 for adopter observations during years 2000 and 2001, 0 otherwise Coded 1 for adopter observations from 2002 to the earlier of 2006 and the year before first ICFR audit 0 otherwise
	ADOPT404a	Coded 1 for adopter observations during year 2007 to the year before first ICFR audit, 0 otherwise
	ADOPT404b	Coded 1 for adopter observations during or after the year of first ICFR audit and during or before the last ICFR audit. 0 otherwise
	AFTERDROP	Coded 1 for adopter observations after ICFR audit is discontinued, 0 otherwise
	BIGN MERGER	Coded 1 if the sample firm is a client of Big N auditor, 0 otherwise Coded 1 if the sample firm has engaged in a merger or acquisition (Compustat footnote AFTNT1), 0 otherwise
	LITIGATION	Coded 1 if the sample firm operates in a high-litigation industry (SIC codes 2833-2836, 3570-3577, 3600-3674, 5200-5961, and 7370), 0 otherwise
	LOSS	Coded 1 if the sample firm reports a loss, 0 otherwise
	L1ACCRUAL	Last year's total accruals equal to net income before extraordinary items (Compustat data item 123) minus cash flow from operations (Compustat data item 308) scaled by lagged total assets
	LEVERAGE	Total asset (Compustat data item 6) less book value (Compustat data item 60) divided by total asset
	MB	Market-to-book ratio
	LNMVE	Natural logarithm of a firm's market value of equity, calculated as price per share at fiscal year-end (Compustat data item 199) multiplied by the number of shares outstanding (Compustat data item 25) (millions of dollars)
	CFO SDCFO	Cash flow from operations (Compustat data item 308) scaled by lagged total assets Standard deviation of the ratio of CFO to total assets over the past five years (including current year)
	SDREV	Standard deviation of the ratio of sales (Compustat data item 12) to total assets over the past five years (including current year)
	∆SALES ASSETS	Change in sales (Compustat data item 12) scaled by lagged total assets Total assets (Compustat data item 6)
<i>T</i> 11 4 T	ΔAR	Change in accounts receivable (Compustat data item 2) scaled by lagged total assets
Table AI. Variable definitions	PPE	Net property, plant and equipment (Compustat data item 8) scaled by lagged total assets

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com



Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

